Author's response to reviews

Title: Documentation of best interest by intensivists: a retrospective study in an Ontario critical care unit

Authors:

Mohana Ratnapalan (mohana.ratnapalan@utoronto.ca)
Andrew B Cooper (andrew.cooper@williamoslerhs.ca)
Damon C Scales (damon.scales@sunnybrook.ca)
Ruxandra Pinto (ruxandra.pinto@sunnybrook.ca)

Version: 5 Date: 5 February 2010

Author's response to reviews:

have made one correction on paragraph 3 of the discussion: the text should read "health care consent act". the word "canada" has been deleted before this.